Author's response to reviews

Title: Assessment of phytochemical content, polyphenolic composition, antioxidant and antibacterial activities of Leguminosae medicinal plants in Peninsular Malaysia

Authors:

Yik Ling Chew (yikling82@yahoo.com)
Wan Ling Elaine Chan (elainechan87@yahoo.com)
Pei Ling Tan (ashleytan23@gmail.com)
Joo Kheng Goh (goh.joo.kheng@sci.monash.edu.my)
Yau Yan Lim (Lim.Yau.Yan@sci.monash.edu.my)
Johnson Stanlas (rcxjs@medic.upm.edu.my)

Version: 3  Date: 11 January 2011

Author's response to reviews: see over
Author’s response to reviews

Title: Assessment of phytochemical content, polyphenolic composition, antioxidant and antibacterial activities of Leguminosae medicinal plants in Peninsular Malaysia

Authors:

Yik-Ling Chew: yikling82@yahoo.com

Elaine-Wan-Ling Chan: elainechan87@yahoo.com

Pei-Ling Tan: ashleytan23@gmail.com

Yau-Yan Lim: Lim.Yau.Yan@sci.monash.edu.my

Johnson Stanslas: rcxjs@edic.upm.edu.my; jstanslas@yahoo.co.uk

Joo-Kheng Goh: goh.joo.kheng@sci.monash.edu.my

Version: 3 Date: 11 January 2011

Author’s response to reviews: Please see over.
Dear Editor,

Please find below the authors’ responses to the reviewer’s comments. The manuscript has been revised according to the comments. I hope the manuscript is now acceptable for publication.

Yours sincerely,

JK Goh

Response to Reviewer’s comments:

**Reviewer:** Shela Gorinstein

**Reviewer's report:**

I confirm that the authors did the corrections.

I suggest to improve the English style in the paper, then to reduce the amount of the references and to include at least one Figure in order to make the paper more attractive.

After these comments you can accept from my side the paper and the other reviewers wrote just wonderful reviews.

*Response:* The quality of written English has been checked and improved. The manuscript was reviewed and the number of references has been reduced from 57 to 50 references. One Figure (Figure 1) reporting on TPC values has been added to make the manuscript attractive.